Author's response to reviews

Title: Predictors of Mortality among TB-HIV Co-infected Patients being treated for Tuberculosis in Northwest Ethiopia: A retrospective cohort study

Authors:

Balewgizie Sileshi (bg.sileshi@gmail.com)
Nigussie Deyessa (negdaysun@yahoo.com)
Belaineh Girma (bgirma@msh.org)
Muluken Melese (mmelese@msh.org)

Version: 2 Date: 24 December 2012

Author's response to reviews: see over
Cover letter

Here in resource limited country evidence based interventions are vital. Like other Sub Saharan countries, Ethiopia is highly affected by the burden of tuberculosis and HIV/AIDS. Most of the time studies conducted remained under shelf which makes difficult to access the result. Therefore, we decided to submit the manuscript and it can be accessible easily for policy planners and researchers. While searching different international journals, we found that BMC is important in many dimensions to submit the manuscript. There are no competing interests. We recommend Wakgare Deressa to be one of the reviewers of the manuscript.

Address of suggested reviewer

Wakgari Deressa, (MD, MPH, PhD), Assistant Professor of Epidemiology and Public Health, School of Public Health, Addis Ababa University.

Phone; +251 911483714

E-mail; deressaw@yahoo.com